# UCSF UC San Francisco Previously Published Works

# Title

Mass azithromycin distribution for hyperendemic trachoma following a cluster-randomized trial: A continuation study of randomly reassigned subclusters (TANA II)

# Permalink

https://escholarship.org/uc/item/47x8q048

**Journal** PLOS Medicine, 15(8)

ISSN

1549-1277

## Authors

Keenan, Jeremy D
Tadesse, Zerihun
Gebresillasie, Sintayehu
et al.

# **Publication Date**

2018

# DOI

10.1371/journal.pmed.1002633

# **Copyright Information**

This work is made available under the terms of a Creative Commons Attribution License, available at <u>https://creativecommons.org/licenses/by/4.0/</u>

Peer reviewed



# 

**Citation:** Keenan JD, Tadesse Z, Gebresillasie S, Shiferaw A, Zerihun M, Emerson PM, et al. (2018) Mass azithromycin distribution for hyperendemic trachoma following a cluster-randomized trial: A continuation study of randomly reassigned subclusters (TANA II). PLoS Med 15(8): e1002633. https://doi.org/10.1371/journal.pmed.1002633

**Academic Editor:** Zulfiqar A. Bhutta, The Hospital for Sick Children, CANADA

Received: March 20, 2018

Accepted: July 5, 2018

Published: August 14, 2018

**Copyright:** © 2018 Keenan et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** Data for all analyses presented in this manuscript are available in the main tables and supplemental files. Specifically, ocular chlamydia data can be found in Tables <u>3</u> and <u>4</u>. Active trachoma data can be found in <u>S1</u> and <u>S2</u> Tables.

**Funding:** This work was supported by the National Institutes of Health (NEI U10 EY016214 and NIH/ NEI K23EY019071), the Bernard Osher Foundation, That Man May See, the Harper Inglis Trust, the RESEARCH ARTICLE

# Mass azithromycin distribution for hyperendemic trachoma following a clusterrandomized trial: A continuation study of randomly reassigned subclusters (TANA II)

Jeremy D. Keenan<sup>1,2</sup>\*, Zerihun Tadesse<sup>3</sup>, Sintayehu Gebresillasie<sup>3</sup>, Ayalew Shiferaw<sup>3</sup>, Mulat Zerihun<sup>3</sup>, Paul M. Emerson<sup>4</sup>, Kelly Callahan<sup>4</sup>, Sun Y. Cotter<sup>1</sup>, Nicole E. Stoller<sup>1</sup>, Travis C. Porco<sup>1,2,5</sup>, Catherine E. Oldenburg<sup>1,2,5</sup>, Thomas M. Lietman<sup>1,2,5</sup>

1 Francis I. Proctor Foundation, University of California, San Francisco, San Francisco, California, United States of America, 2 Department of Ophthalmology, University of California, San Francisco, San Francisco, California, United States of America, 3 The Carter Center, Ethiopia, Addis Ababa, Ethiopia, 4 The Carter Center, Atlanta, Georgia, United States of America, 5 Department of Epidemiology and Biostatistics, University of California, San Francisco, San Francisco, California, United States of America, 5 Department of Epidemiology and Biostatistics, University of California, San Francisco, San Francisco, California, United States of America

\* jeremy.keenan@ucsf.edu

# Abstract

## Background

The World Health Organization recommends annual mass azithromycin administration in communities with at least 10% prevalence of trachomatous inflammation–follicular (TF) in children, with further treatment depending on reassessment after 3–5 years. However, the effect of stopping mass azithromycin distribution after multiple rounds of treatment is not well understood. Here, we report the results of a cluster-randomized trial where communities that had received 4 years of treatments were then randomized to continuation or discontinuation of treatment.

## Methods and findings

In all, 48 communities with 3,938 children aged 0–9 years at baseline in northern Ethiopia had received 4 years of annual or twice yearly mass azithromycin distribution as part of the TANA I trial. We randomized these communities to either continuation or discontinuation of treatment. Individuals in the communities in the continuation arm were offered either annual or twice yearly distribution of a single directly observed dose of oral azithromycin. The primary outcome was community prevalence of ocular chlamydial infection in a random sample of children aged 0–9 years, 36 months after baseline. We also assessed the change from baseline to 36 months in ocular chlamydia prevalence within each arm. We compared 36-month ocular chlamydia prevalence in communities randomized to continuation versus discontinuation in a model adjusting for baseline ocular chlamydia prevalence. A secondary prespecified analysis assessed the rate of change over time in ocular chlamydia prevalence between arms. In the continuation arm, mean antibiotic coverage was greater than 90% at all time points. In the discontinuation arm, the mean prevalence of infection in children aged



Bodri Foundation, the South Asia Research Fund, Research to Prevent Blindness, and the International Trachoma Initiative (through a donation of azithromycin). The trachoma control program in Amhara is supported by the The Carter Center, Amhara Regional Health Bureau, the Lions-Carter Center, SightFirst Initiative and many individual donors. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

**Abbreviations:** TF, trachomatous inflammation–follicular.

0–9 years increased from 8.3% (95% CI 4.2% to 12.4%) at 0 months to 14.7% (95% CI 8.7% to 20.8%, P = 0.04) at 36 months. Ocular chlamydia prevalence in communities where mass azithromycin distribution was continued was 7.2% (95% CI 3.3% to 11.0%) at baseline and 6.6% (95% CI 1.1% to 12.0%, P = 0.64) at 36 months. The 36-month prevalence of ocular chlamydia was significantly lower in communities continuing treatment compared with those discontinuing treatment (P = 0.03). Limitations of the study include uncertain generalizability outside of trachoma hyperendemic regions.

## Conclusions

In this study, ocular chlamydia infection rebounded after 4 years of periodic mass azithromycin distribution. Continued distributions did not completely eliminate infection in all communities or meet WHO control goals, although they did prevent resurgence.

## **Trial registration**

This study was prospectively registered at clinicaltrials.gov (clinicaltrials.gov NCT01202331).

## Author summary

## Why was this study done?

- Trachoma is the leading infectious cause of blindness in the world.
- Although trachoma is disappearing globally, severe trachoma persists in a few districts in Ethiopia.
- Periodic mass azithromycin distribution reduces infection to very low levels, but complete elimination in all communities has proven difficult.

## What did the researchers do and find?

- We assessed whether mass azithromycin distributions could be discontinued after 4 years of treatment in communities with severe trachoma in northern Ethiopia.
- A subset of communities that had participated in a previous cluster-randomized trial were re-randomized to continuation or discontinuation of mass azithromycin distribution.
- Ocular chlamydia infection, which causes trachoma, was found to increase over time after mass azithromycin distribution was stopped. Compared to communities where mass azithromycin distribution was continued, communities where distribution was stopped had significantly higher rates of ocular chlamydia infection after 3 years of study.
- Despite 7 years of treatment in communities continuing mass azithromycin distribution, infection was not eliminated in all communities.

## What do these findings mean?

- Trachoma may be the first bacterial disease of humans to be eliminated by a public health program; however, these results indicate that stopping mass azithromycin treatment in some severely affected areas is not realistic.
- Alternative strategies for trachoma elimination may be required for the most severely affected areas.

## Introduction

Mass distribution of azithromycin is a key component of the World Health Organization's SAFE (surgery for trichiasis, antibiotics, facial cleanliness, and environmental improvement) strategy for trachoma control [1–3]. Antibiotics (azithromycin), facial cleanliness, and environmental improvements (e.g., latrinization and water improvements) are interventions aimed at reducing community transmission of the strains of ocular chlamydia that cause trachoma [4,5]. Although facial cleanliness and environmental improvements are associated with decreased clinical trachoma, we have no evidence that facial hygiene or environmental improvements have any effect on ocular chlamydia [6-8]. On the other hand, a single mass azithromycin distribution has been shown to dramatically reduce the prevalence of ocular chlamydia [9,10]. The WHO currently recommends 3 to 5 rounds of annual mass azithromycin distributions for communities in which the district prevalence of trachomatous inflammation-follicular (TF) among 1- to 9-year-old children exceeds 10%, followed by an impact survey to assess whether treatment should be continued. If the district prevalence of TF is below 10% during the impact survey, then community-level surveys should be done, and any communities with a prevalence of TF < 5% should have mass antibiotic treatments discontinued [11].

The WHO recommendation for when to continue mass azithromycin distribution, while extremely valuable to trachoma programs, is based on very limited empirical evidence. Some studies have found evidence to support the existence of an Allee-like effect, in which chlamydial infections fade away if reduced below some low threshold [9,12–15]. In contrast, other studies have found that infection rapidly rebounds once repeated mass azithromycin distributions are stopped [16,17].

In a previous trial (TANA I), we compared repeated annual versus biannual mass azithromycin distributions to determine whether more frequent azithromycin treatment would lead to better control of ocular chlamydia [18]. The trial found a marked reduction in the prevalence of ocular chlamydia among 0- to 9-year-olds in both the annual and biannual groups over the duration of the 3.5-year study, but no significant difference between the 2 treatment strategies at the final study visit [18]. Following the completion of this study, we initiated a continuation trial (TANA II) in which communities from the annual and biannual arms of the original trial were re-randomized to continuing or stopping mass azithromycin distribution to evaluate the need for continued mass azithromycin distributions in areas that have received multiple rounds of mass azithromycin distribution. Here, we report the effect of continuation versus discontinuation of mass azithromycin treatment following 4 years of repeated annual or biannual mass azithromycin distribution on ocular chlamydial infection in a hyperendemic region of Ethiopia.

#### Methods

TANA (Trachoma Amelioration in Northern Amhara, ClinicalTrials.gov NCT00322972) was a community-randomized trial that compared different mass azithromycin strategies for reducing ocular chlamydial infection in a hyperendemic area of Ethiopia. The present report describes 4 arms from its continuation trial, Tripartite International Research for the Elimination of Trachoma (TIRET) (ClinicalTrials.gov NCT01202331), also known as and referred to here as TANA II. The trial received ethical approval from the University of California, San Francisco; Emory University; and the Ethiopian Ministry of Science and Technology. Because illiteracy was common among study participants, we obtained verbal informed consent for all study procedures. Caregivers provided verbal consent for children to participate, and community leaders provided consent for their community to participate in the study.

#### **TANA I overview**

TANA I was a cluster-randomized trial conducted in the Goncha Siso Enese woreda (district) of the Amhara Region of Ethiopia from June 1, 2006, to December 2, 2009 [18]. All communities in the district were eligible for randomization excluding the town of Gindewoin, as it is a semi-urban area with a lower prevalence of trachoma. In the trial, we randomized 72 government-defined subdivisions known as subkebeles to 1 of 6 different treatment strategies, including 12 subkebeles to annual mass azithromycin treatments and 12 subkebeles to biannual mass azithromycin treatments (Fig 1). Subkebeles consisted of 4–6 government-defined units known as state teams (communities). We treated all communities of the subkebele identically, and performed monitoring for ocular chlamydia in a single sentinel state team from the subkebele. We collected conjunctival swabs every 6 months from a random sample of community members, and processed swabs for chlamydia testing using AMPLICOR CT (Roche Molecular Diagnostics, Pleasanton, CA).

## **TANA II overview**

We conducted the TANA II continuation trial from November 1, 2010, until November 8, 2013. The TANA II trial population consisted of 48 state teams that had received annual mass azithromycin distributions and 24 state teams that had received biannual mass azithromycin distributions during TANA I. State teams were randomly selected stratified by subkebele, with 4 state teams from each of the 12 annually treated subkebeles and 2 state teams from each of the 12 biannually treated subkebeles. We randomized the 4 state teams in each annually treated subkebele to 1 of the following treatment strategies: (1) discontinuation of mass treatments, (2) continued annual mass azithromycin treatments, (3) azithromycin treatments targeted to preschool children, or (4) azithromycin treatments targeted to households containing children with clinically active trachoma. We randomized the 2 state teams from each biannually treated subkebele to either discontinuation of mass treatments or to continued biannual mass azithromycin distributions. As depicted in Fig 1, the present report includes 48 state teams randomized into 4 treatment groups: continuation of annual mass azithromycin (N = 12), cessation of annual mass azithromycin (N = 12), continuation of biannual mass azithromycin (N = 12), and cessation of biannual mass azithromycin (N = 12). The 2-stage randomization means that these 4 groups were randomly drawn from the same underlying population and can thus be validly compared with each other. The trial protocol and outcomes were prespecified, and there were no changes to the methods after commencement.

#### Randomization and masking

TCP performed the randomization in the statistical package R (R Foundation for Statistical Computing, Vienna, Austria). State teams from the annual and biannual arms of TANA I were



Fig 1. CONSORT diagram. Age range for children is 0-9 years; age range for adults is 10 years and older.

https://doi.org/10.1371/journal.pmed.1002633.g001

randomized to discontinue or continue their treatment strategy. All state teams were enrolled prior to randomization. Study participants were not masked to their state team's treatment allocation. Field staff were not informed of, and laboratory personnel were masked to, the state team's treatment allocation.

### Intervention

In both TANA I and TANA II, we performed an annual door-to-door enumerative census. In state teams assigned to treatment, azithromycin was offered to all community members aged 6 months and up (approximately 20 mg/kg using height-based dosing for children, up to a maximum of 1 g) [19]. Children less than 6 months, pregnant women, and those allergic to macro-lide antibiotics were offered 2 tubes of topical tetracycline instead, to be used twice daily for 6 weeks. Antibiotic coverage was defined as the proportion of individuals from the most recent census who received either oral azithromycin or topical tetracycline. An adverse event notification system was put in place for individuals to notify the village informant and the Ethiopian study coordinator.

#### Monitoring

The primary outcome was prevalence of ocular chlamydia infection as detected by PCR. Active trachoma was a prespecified secondary outcome and is included in this report as it is an important component of monitoring trachoma control programs [20]. We performed annual assessments in the same populations as in TANA I: a random sample of 50 children aged 0–9 years and a random sample of 50 individuals aged 10 years and older ("adults") from each state team based on the most recent census. To assess active trachoma, at each monitoring visit, we examined the everted upper right tarsal conjunctiva for signs of trachoma according to the WHO simplified trachoma grading system, specifically assessing the presence of TF and trachomatous inflammation–intense. To assess ocular chlamydia, we then collected a conjunctival specimen by passing a Dacron swab 3 times against the everted tarsal conjunctiva, rotating the swab 120° between each pass. We stored swabs on ice for <8 hours while in the field, then at -20°C until transport to the lab, at 4°C for <24 hours during transport, and finally at -80°C until processing. We performed chlamydial testing with the Abbott m2000 platform on pools of 5 randomly selected swabs, with pools stratified by community and age group. We estimated the prevalence of infection most likely to have resulted in the pooled result, as previously described [21].

#### Sample size determination

The sample size was based on the primary outcome for the discontinuation and continuation arms, with assumptions based on the results of previous studies (the Trachoma Elimination Follow-up study and TANA I) [18,22]. The standard deviation after 1 year of treatment in TANA I was 0.043, and we conservatively assumed a standard deviation of 0.05 for sample size planning. For assessment of rebound of ocular chlamydia following discontinuation of treatment, a sample size of 12 state teams per arm was determined to provide approximately 80% power to detect a difference of 5% assuming a paired *t* test. For comparison of continued annual versus continued biannual treatment, a sample size of 12 state teams per arm was determined to provide approximately 80% power to detect a 6% difference in ocular chlamydia prevalence.

#### Statistical considerations

All analyses were conducted at the level of the unit of randomization, the state team, to account for the clustered design of the trial. We estimated the prevalence of ocular chlamydia and

active trachoma in each state team at the community level as described above. The prespecified primary outcome was the prevalence of ocular chlamydia in children aged 0–9 years at 36 months after the baseline visit of TANA II. All analyses were conducted in Stata 14.2 (Stata-Corp, College Station, TX).

Longitudinal changes in prevalence. When analyzing the change in prevalence longitudinally, we analyzed prevalence itself (non-transformed). The prespecified primary analysis was the difference in prevalence from TANA II baseline to 36 months in the discontinuation arm, with communities that had had annual and biannual azithromycin distribution in TANA I aggregated together. In communities randomized to stopping mass azithromycin distribution, we first tested whether the prevalence of ocular chlamydia at 36 months was different from that at the baseline TANA II visit using a paired *t* test. As a prespecified secondary analysis, we modeled chlamydia prevalence over all 4 study visits using a mixed effects linear regression model in which the study community and its study visit were designated as random effects (allowing for a random slope and random intercept for each community's infection results over time), with treatment arm as a fixed effect. As a secondary analysis, we repeated the analysis with a Wilcoxon signed-rank test. A similar analytic strategy was used to assess the prevalence of ocular chlamydia over time in the continuation arm.

**Comparison of continuation versus discontinuation.** In analyses comparing the 2 arms, we square-root-transformed the prevalence to improve the model fit. Because communities were re-randomized, we included TANA II baseline prevalence as a covariate in models of continuation versus discontinuation of treatment. The prespecified analysis compared the prevalence of infection at 36 months (i.e., 90 months after randomization in TANA I) between the discontinuation and continuation arms in a multiple linear regression with a term for TANA II baseline ocular chlamydia prevalence. As a prespecified sensitivity analysis, we included data from all 4 TANA II monitoring visits in a mixed effects linear regression in which the study community and its study visit were designated as random effects, allowing for a random slope and random intercept for each community's infection results over time, with treatment arm as a fixed effect. A Monte Carlo permutation test stratified by TANA I study arm with 10,000 replications was used to calculate the *P*-value. As a sensitivity analysis, we compared ocular chlamydia in communities in the continuation and discontinuation arms separately by antibiotic distribution schedule (e.g., annual versus biannual), with an interaction term between TANA I study arm and continuation versus discontinuation. A similar analytic approach was used for TF outcomes.

**Comparison of annual versus biannual treatment.** To compare the results of communities continuing annual versus continuing biannual mass azithromycin distribution, we used a multiple linear regression model with terms for the TANA I baseline value and distribution schedule. Because communities were not re-randomized to annual versus biannual treatment at the start of TANA II, we included the TANA I baseline value: inclusion of the TANA II baseline value would have been a post-randomization correction.

## Results

During the first TANA trial, 12 subkebeles received 4 annual mass azithromycin distributions and 12 subkebeles received 8 biannual mass azithromycin distributions over 3 years, with antibiotic coverage generally exceeding 80% in each community [18]. The results of the TANA I trial have been reported previously [18]. Briefly, the prevalence of ocular chlamydial infection in 0- to 9-year-olds as assessed by AMPLICOR decreased in both of the treatment groups after 3 years of repeated mass azithromycin distributions, from 41.9% to 1.9% in the annual group and from 38.3% to 3.2% in the biannual group. The baseline visit for TANA II took place 54

Characteristic	Discontinuation arm		Continuation arm	
	Annual treatment	Biannual treatment	Annual treatment	Biannual treatment
Number of state teams	12	12	12	12
Population <sup>1</sup>				
0–9 years	360 (294–430)	274 (222–330)	300 (246–354)	330 (276–391)
$\geq 10$ years	90 (73–107)	75 (59–92)	81 (63–98)	93 (77–109)
Proportion female <sup>2</sup>				
0–9 years	51.2% (48.2% to 53.9%)	49.5% (45.1% to 53.7%)	48.5% (45.7% to 51.7%)	49.3% (46.7% to 52.1%)
$\geq 10$ years	50.2% (48.6% to 51.8%)	53.2% (51.2% to 55.6%)	51.3% (49.5% to 53.0%)	50.8% (49.0% to 52.6%)
Proportion with TF/TI <sup>2</sup>				
0–9 years	31.4% (22.3% to 39.8%)	34.5% (23.5% to 46.2%)	42.0% (34.7% to 49.0%)	35.9% (22.7% to 47.8%)
$\geq 10$ years	4.6% (1.9% to 9.2%)	4.2% (2.2% to 6.5%)	3.9% (2.2% to 6.1%)	2.9% (1.5% to 5.3%)
Elevation (m) <sup>2</sup>	2,585.5 (2,456.3 to 2,679.0)	2,606.4 (2,466.2 to 2,721.2)	2,510.3 (2,354.3 to 2,638.3)	2,531.2 (2,374.2 to 2,682.7)

#### Table 1. Baseline characteristics by study arm.

TF, trachomatous inflammation-follicular; TI, trachomatous inflammation-intense.

<sup>1</sup>Range given in parentheses.

<sup>2</sup>95% confidence interval given in parentheses.

https://doi.org/10.1371/journal.pmed.1002633.t001

months after the baseline TANA I visit, which was 18 months after the final mass azithromycin distribution in the annually treated group, and 12 months after the final mass azithromycin distribution in the biannually treated group. Community characteristics for the baseline TANA II visit are shown in Table 1. Antibiotic coverage during TANA II for the 24 communities randomized to continuation of annual or biannual mass azithromycin distribution generally exceeded 80% at each distribution (Table 2). No adverse events were reported in the study.

Tables 3 and 4 show the prevalence of ocular chlamydia among children aged 0–9 years in each state team over time in the discontinuation and continuation arms, respectively. In the 24 state teams in which mass azithromycin distribution was discontinued, the mean prevalence of ocular chlamydia infection increased from 8.3% (95% CI 4.2% to 12.4%) at TANA II baseline to 14.7% (95% CI 8.7% to 20.8%) at the 36-month visit (P = 0.04, paired t test). This result was consistent with a repeated measures model that included all time points (P = 0.01) and with a Wilcoxon signed-rank test (P = 0.054). In contrast, the prevalence of ocular chlamydia did not change markedly in the 24 state teams where mass azithromycin distribution was continued, decreasing from 7.2% (95% CI 3.3% to 11.0%) at TANA II baseline to 6.6% (95% CI 1.1% to 12.0%) at 36 months (P = 0.85, paired t test), although a repeated measures model including all time points indicated a decrease in prevalence over time (P = 0.047).

Table 2. A	Antibiotic	coverage b	y study	arm.
------------	------------	------------	---------	------

Time point	Discontinuation arm	Discontinuation arm		Continuation arm	
	Annual treatment	<b>Biannual treatment</b>	Annual treatment <sup>1</sup>	Biannual treatment <sup>1</sup>	
0 months	_	_	93.7% (90.6% to 96.3%)	90.5% (87.9% to 93.4%)	
6 months	-	-	—	91.0% (88.3% to 93.4%)	
12 months	_	_	92.9% (88.7% to 96.7%)	91.4% (88.5% to 94.3%)	
18 months	_	_	_	92.8% (90.7% to 95.0%)	
24 months	-	-	95.8% (94.3% to 97.2%)	94.3% (91.7% to 96.7%)	
30 months	_	_	_	95.0% (93.5% to 96.5%)	
36 months	_	_	91.9% (87.3% to 94.3%)	91.7% (87.8% to 94.7%)	

<sup>1</sup>95% confidence interval given in parentheses.

https://doi.org/10.1371/journal.pmed.1002633.t002

State team	Prevalence of ocular chlamydia, 0- to 9-year-old children				
	0 months	12 months	24 months	36 months	
Annual treatme	ent				
1	2.0% (1/50)	8.9% (5/56)	3.8% (2/53)	13.9% (5/36)	
2	5.7% (3/53)	6.0% (3/50)	15.1% (8/53)	16.0% (8/50)	
3	12.7% (7/55)	6.1% (3/49)	18.5%)(10/54)	3.7% (2/54)	
4	22.6% (12/53)	21.8% (12/55)	12.5% (6/48)	17.5% (7/40)	
5	0.0% (0/46)	8.9% (4/45)	13.0% (7/54)	24.5% (12/49)	
6	0.0% (0/51)	0.0% (0/43)	0.0% (0/44)	16.7% (8/48)	
7	5.8% (3/52)	0.0% (0/37)	4.3% (2/46)	2.0% (1/50)	
8	7.8%(4/51)	2.1% (1/47)	6.1% (3/49)	3.9% (2/51)	
9	0.0% (0/26)	0.0% (0/28)	0.0% (0/26)	0.0% (0/21)	
10	0.0% (0/52)	0.0% (0/56)	10.9% (6/55)	17.8% (8/45)	
11	36.0% (18/50)	34.0% (16/47)	41.7% (20/48)	46.8% (22/47)	
12	26.0% (13/50)	11.8% (6/51)	16.3% (8/49)	4.2% (2/48)	
Mean (SD)	9.9% (12.1)	8.3% (10.4)	11.9% (11.3)	13.9% (13.0)	
Biannual treat	nent				
13	20.0% (8/40)	16.2% (6/37)	34.2% (13/38)	20.0% (8/40)	
14	2.3% (1/44)	0.0% (0/43)	0.0% (0/36)	0.0% (0/37)	
15	2.0% (1/50)	18.2% (10/55)	26.5% (13/49)	33.3% (17/51)	
16	7.8% (4/51)	5.9% (3/51)	20.4% (10/49)	4.0% (2/50)	
17	17.4% (8/46)	17.1% (7/41)	27.1% (13/48)	23.3% (10/43)	
18	11.8% (6/51)	11.8% (6/51)	27.9% (12/43)	28.9% (11/38)	
19	0.0% (0/49)	1.9% (1/52)	2.0% (1/50)	2.0% (1/50)	
20	8.8% (3/34)	2.9% (1/35)	0.0% (0/32)	0.0% (0/27)	
21	0.0% (0/48)	0.0% (0/52)	0.0% (0/55)	0.0% (0/49)	
22	2.0% (1/49)	0.0% (0/44)	2.0% (1/51)	37.3% (19/51)	
23	8.0% (4/50)	22.9% (11/48)	30.8% (16/52)	37.5% (18/48)	
24	0.0% (0/40)	0.0% (0/34)	0.0% (0/45)	0.0% (0/41)	
Mean (SD)	6.7% (6.9)	8.1% (8.6)	14.2% (14.5)	15.5% (16.0)	

Table 3. Longitudinal prevalence of ocular chlamydia among a random sample of 0- to 9-year-old children after discontinuation of mass azithromycin distribution.

https://doi.org/10.1371/journal.pmed.1002633.t003

At 36 months, there was no significant difference between state teams that had continued to receive annual (9.9%, 95% CI 0% to 20.4%) versus biannual (3.3%, 95% CI 0% to 7.5%) mass azithromycin distributions (P = 0.09, linear regression adjusted for TANA I baseline infection prevalence).

The 24 state teams randomized to discontinuation of mass azithromycin distributions had a significantly higher prevalence of ocular chlamydia at 36 months (14.7%, 95% CI 8.7% to 20.8%) compared to the 24 state teams that continued treatment (6.6%, 95% CI 1.1% to 12.0%; P = 0.03, linear regression adjusted for prevalence of infection at TANA II baseline). In a prespecified sensitivity analysis including data from all time points in a repeated measures model with a time by treatment arm interaction term, state teams randomized to discontinuation had significantly higher ocular chlamydia prevalence over time compared to those that continued mass azithromycin distributions (*P*-interaction = 0.02; Fig 2). There was no evidence of effect modification by treatment frequency (annual or biannual) of the effect of continuation versus discontinuation (*P*-interaction = 0.43). However, the reduction in prevalence in the continuation compared to the discontinuation arm was greater in biannually treated communities compared to annually treated communities (P = 0.04 biannual, P = 0.30 annual).

State team	Prevalence of ocular chlamydia, 0- to 9-year-old children				
	0 months	12 months	24 months	36 months	
Annual treatme	ent				
25	0.0% (0/32)	5.6% (2/36)	5.9% (2/34)	5.3% (2/38)	
26	10.0% (3/30)	0.0% (0/26)	0.0% (0/33)	0.0% (0/29)	
27	11.1% (6/54)	2.1% (1/47)	4.8% (2/42)	2.4% (1/41)	
28	11.1% (0/51)	16.7% (8/48)	0.0% (0/46)	5.7% (3/53)	
29	0.0% (0/50)	0.0% (0/52)	0.0% (0/49)	20.4% (10/49)	
30	16.7% (8/48)	0.0% (0/49)	1.9% (1/52)	3.6% (2/55)	
31	8.0% (4/50)	0.0% (0/47)	0.0% (0/48)	1.9% (1/54)	
32	3.7% (2/54)	2.0% (1/49)	0.0% (0/49)	2.5% (1/40)	
33	0.0% (0/33)	0.0% (0/38)	2.4% (1/42)	0.0% (0/38)	
34	1.9% (1/52)	0.0% (0/52)	0.0% (0/53)	5.8% (3/52)	
35	11.8% (6/51)	6.0% (3/50)	7.7% (4/52)	59.5% (25/42)	
36	30.8% (16/52)	11.5% (6/52)	3.9% (2/51)	11.3% (6/53)	
Mean (SD)	8.8% (8.9)	3.7% (5.4)	2.2% (2.7)	9.9% (16.6)	
Biannual treatm	nent				
37	23.4% (11/47)	5.0% (2/40)	7.5% (3/40)	0.0% (0/38)	
38	0.0% (0/45)	4.8% (2/42)	0.0% (0/39)	0.0% (0/40)	
39	3.6% (2/55)	1.9% (1/53)	0.0% (0/57)	0.0% (0/52)	
40	0.0% (0/52)	0.0% (0/52)	0.0% (0/50)	0.0% (0/38)	
41	27.7% (13/47)	2.1% (1/48)	20.0% (10/50)	1.9% (1/54)	
42	4.3% (2/47)	12.7% (7/55)	3.8% (2/52)	23.4% (11/47)	
43	0.0% (0/51)	0.0% (0/53)	0.0% (0/55)	0.0% (0/52)	
44	0.0% (0/48)	0.0% (0/38)	0.0% (0/37)	6.3% (2/32)	
45	0.0% (0/53)	0.0% (0/52)	0.0% (0/49)	0.0% (0/56)	
46	4.0% (2/50)	0.0% (0/49)	0.0% (0/52)	5.8% (3/52)	
47	4.0% (2/50)	4.0% (2/50)	8.5% (4/47)	1.9% (1/52)	
48	0.0% (0/50)	0.0% (0/51)	0.0% (0/51)	0.0% (0/53)	
Mean (SD)	5.6% (9.5)	2.5% (3.8)	3.3% (6.1%)	3.3% (6.7)	

Table 4. Longitudinal prevalence of ocular chlamydia among a random sample of 0- to 9-year-old children with continuation of mass azithromycin distribution.

https://doi.org/10.1371/journal.pmed.1002633.t004

Similar analyses were performed for TF in order to assess changes in clinically active trachoma (S1 and S2 Tables). In state teams in which mass azithromycin distribution was discontinued, the mean prevalence of TF at TANA II baseline was 31.1% (95% CI 23.9% to 38.3%), which increased to 39.9% (95% CI 32.5% to 47.4%) at 36 months (P = 0.02, paired t test). In communities in which mass azithromycin distributions were continued, the mean TF prevalence at TANA II baseline was 37.9% (95% CI 30.0% to 45.8%), which decreased to 32.6% (95% CI 24.8% to 40.4%) at 36 months (P = 0.05, paired t test). Communities in which mass azithromycin distributions were discontinued had significantly higher TF prevalence at 36 months compared to communities in which mass azithromycin distributions were continued (mean difference 11.7%, 95% CI 3.5% to 20.0%; P = 0.004, linear regression model adjusted for TF prevalence at TANA II baseline). In a repeated measures sensitivity analysis, state teams randomized to discontinuation of mass azithromycin distribution had significantly higher TF prevalence over time (time by treatment P-interaction = 0.006, mixed effects model; Fig 3). There was no significant difference in change in TF prevalence in communities originally randomized to continuation versus discontinuation in annually and biannually treated communities (*P*-interaction = 0.07), although, similar to ocular chlamydia prevalence, there was a



Fig 2. Longitudinal prevalence of ocular chlamydia in children aged 0–9 years. Black line is the mean in all communities. Grey lines are the prevalence in each community.

https://doi.org/10.1371/journal.pmed.1002633.g002

PLOS MEDICINE

stronger effect of continuation in biannually compared to annually treated communities (P = 0.001 biannual, P = 0.41 annual).

### Discussion

Although continuation of mass azithromycin distribution led to reduced prevalence of ocular chlamydia compared to communities in which distribution was discontinued, this study demonstrates that even 7 rounds of mass azithromycin were not sufficient for elimination in an area with hyperendemic trachoma. We initially hypothesized that several rounds of mass azi-thromycin distributions might completely eliminate ocular chlamydia [18]. Previous work with topical tetracycline ointment indicated that antibiotics alone would not be sufficient to achieve trachoma elimination [23]. However, limitations of topical tetracycline, including long treatment duration and poor adherence, may reduce its effectiveness [24]. Single-dose azithromycin can overcome many of these limitations, although previous work has indicated ongoing





Fig 3. Longitudinal prevalence of trachomatous inflammation-follicular in children aged 0–9 years. Black line is the mean in all communities. Grey lines are the prevalence in each community.

https://doi.org/10.1371/journal.pmed.1002633.g003

transmission of ocular chlamydia even after 3 years of implementation of the SAFE strategy in a hyperendemic region of Ethiopia [25]. In the present study, in communities that had previously received 4 years of repeated mass azithromycin distributions, the prevalence of both clinically active trachoma and ocular chlamydia infection stabilized if mass treatments were continued, and increased if mass treatments were stopped. Continuing antibiotic distributions was superior to discontinuing treatments, arguing for continued rounds of antibiotics if the goal is maintaining low levels of ocular chlamydia.

Communities in which treatment was discontinued demonstrated a slower rate of return of infection than previous studies in other hyperendemic regions of Ethiopia. A study in the Gurage zone of Ethiopia found that the average prevalence of ocular chlamydia among 1- to 5-year-old children decreased from 64% to 2.6% after 4 biannual treatments, but returned to 25% 2 years later [26]. In contrast, the communities in the present study experienced slightly less than a 2-fold increase in ocular chlamydia reinfection over 3 years. Several factors may

explain these results. Communities in the present study were treated with mass azithromycin distributions for an additional year, which may have resulted in more sustained reduction of infection. Clusters of communities in the same district were initially treated identically in the present study, which would have reduced infection in neighboring communities and could have slowed reemergent infection. We cannot rule out the possibility of differences in the underlying strength of transmission between the Gurage zone and the Amhara Region. As has been seen in other longitudinal assessments, the correlation of infection prevalence over time is low at lower levels of infection [16,27], and thus there was noise in trends in individual community prevalence. Finally, the presence of a secular trend could have reduced reinfection independently from the mass azithromycin treatments.

Mathematical models have suggested that biannual mass distributions might be sufficient to completely eliminate infection, even in severely affected areas where annual treatment is not [5,9,10,28,29]. We did not find that to be the case in this setting. In approximately one-half of the communities, we were able to identify infection from a sample of children even after 7 years of mass treatments. Several possibilities could explain the failure to eliminate trachoma. Infection could have been introduced from people not included in the control program, including those living in neighboring districts or those in the study district who did not participate in the program. Alternatively, relatively poor water and sanitation conditions may have promoted transmission [4,30,31]. Another possibility is that in this hyperendemic setting, even biannual treatment is not enough to completely eliminate infection from those most likely to transmit infection (e.g., preschool children) and that more intensive treatments are necessary specifically for these hyper-transmitters in the community.

Longer-term distribution of mass azithromycin may increase selection for macrolide resistance. Although previous studies have not demonstrated an increase in resistance selection in *Chlamydia trachomatis*, evidence is limited [32,33]. If multiple years of mass azithromycin distribution increase the probability of resistance selection in *C. trachomatis*, the efficacy of mass azithromycin distribution may wane over time. Mass distribution of azithromycin for trachoma control has previously been shown to lead to selection for macrolide resistance in nontarget organisms including *Streptococcus pneumoniae* and *Escherichia coli*, and resistance declines following cessation of antibiotics [34–37]. Monitoring of macrolide resistance selection in communities receiving long-term mass azithromycin distribution may be important for understanding if multiple rounds of treatment have adverse effects on treatment efficacy.

The results of this study must be considered in the context of several limitations. First, in all communities, infection increased from the end of TANA I until the beginning of TANA II, perhaps in part because of an approximately 6-month lag between the end of TANA I and the beginning of TANA II. Second, this study focused on evaluation of continuation versus discontinuation of mass azithromycin, and did not evaluate combined strategies such as the efficacy of azithromycin in the presence of the facial cleanliness and environmental improvement components of the SAFE strategy. In the study communities, facial cleanliness and environmental improvement initiatives were implemented by the Ethiopian government, but we did not monitor their implementation or community adherence to these components. Note that while facial cleanliness and environmental conditions have been associated with decreases in clinically active trachoma, we have no evidence that facial hygiene or environmental programs have any effect on ocular chlamydial infection [38]. Ongoing work is evaluating the efficacy of facial cleanliness and environmental improvement for trachoma control (ClinicalTrials.gov NCT02754583, Principal Investigator J. Keenan). Third, these results are likely not generalizable to settings with hypo- or mesoendemic trachoma, where infection may be disappearing on its own and may not return after several rounds of mass azithromycin distribution [9,39]. Other hyperendemic areas may have different transmission characteristics, and may thus

respond differently after repeated mass azithromycin distribution. The results are, however, generalizable to similar regions of Ethiopia, which has the highest burden of trachoma in the world and which will provide the world's biggest challenge for trachoma elimination.

Four years of repeated mass azithromycin distributions greatly reduced the prevalence of ocular chlamydia in 0- to 9-year-old children, to approximately 3%. Upon cessation of treatments, infection rebounded. Reducing infection to a low level will not in itself prevent infection from rebounding in this setting. Communities in which annual or biannual mass distributions were continued for an additional 3 years experienced stabilization of infection and clinical disease, suggesting that antibiotics alone may not be enough to completely eliminate infection in severely affected areas. However, continuing mass azithromycin distribution after 3–5 years is significantly better for maintaining low levels of ocular chlamydia than is discontinuing treatment.

## **Supporting information**

**S1 CONSORT Checklist. CONSORT checklist.** (DOC)

**S1** Approvals. Institutional review board approvals for TANA II. (PDF)

**S1 Protocol. Manual of operations and procedures for TANA II.** (PDF)

S1 Table. Longitudinal prevalence of trachomatous inflammation-follicular among a random sample of 0- to 9-year-old children after discontinuation of mass azithromycin distribution.

(DOCX)

S2 Table. Longitudinal prevalence of trachomatous inflammation-follicular among a random sample of 0- to 9-year-old children with continuation of mass azithromycin distribution.

(DOCX)

## Acknowledgments

We would like to thank the Biblioteca Angelica and its staff.

## **Author Contributions**

- **Conceptualization:** Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Paul M. Emerson, Travis C. Porco, Thomas M. Lietman.
- **Data curation:** Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Mulat Zerihun, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Travis C. Porco, Thomas M. Lietman.
- **Formal analysis:** Jeremy D. Keenan, Travis C. Porco, Catherine E. Oldenburg, Thomas M. Lietman.
- **Funding acquisition:** Jeremy D. Keenan, Paul M. Emerson, Nicole E. Stoller, Travis C. Porco, Thomas M. Lietman.

- **Investigation:** Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Mulat Zerihun, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Travis C. Porco, Thomas M. Lietman.
- **Methodology:** Jeremy D. Keenan, Zerihun Tadesse, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Travis C. Porco, Thomas M. Lietman.
- **Project administration:** Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Mulat Zerihun, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Thomas M. Lietman.
- **Resources:** Jeremy D. Keenan, Zerihun Tadesse, Paul M. Emerson, Nicole E. Stoller, Thomas M. Lietman.

Software: Travis C. Porco.

Supervision: Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Mulat Zerihun, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Thomas M. Lietman.

Validation: Travis C. Porco, Catherine E. Oldenburg.

Visualization: Travis C. Porco, Catherine E. Oldenburg.

Writing - original draft: Jeremy D. Keenan, Catherine E. Oldenburg, Thomas M. Lietman.

Writing – review & editing: Jeremy D. Keenan, Zerihun Tadesse, Sintayehu Gebresillasie, Ayalew Shiferaw, Mulat Zerihun, Paul M. Emerson, Kelly Callahan, Sun Y. Cotter, Nicole E. Stoller, Travis C. Porco, Catherine E. Oldenburg.

#### References

- Taylor HR, Burton MJ, Haddad D, West F, Wright H. Trachoma. Lancet. 2014; 384:2142–52. https:// doi.org/10.1016/S0140-6736(13)62182-0 PMID: 25043452
- Mabey DC, Solomon AW, Foster A. Trachoma. Lancet. 2003; 362:223–9. https://doi.org/10.1016/ S0140-6736(03)13914-1 PMID: 12885486
- Hu VH, Harding-Esch EM, Burton MJ, Bailey RL, Kadimpeul J, Mabey DCW. Epidemiology and control of trachoma: systematic review. Trop Med Int Health. 2010; 15:673–91. https://doi.org/10.1111/j.1365-3156.2010.02521.x PMID: 20374566
- Emerson PM, Cairncross S, Bailey RL, Mabey DCW. Review of the evidence base for the 'F' and 'E' components of the SAFE strategy for trachoma control. Trop Med Int Health. 2000; 5:515–27. PMID: 10995092
- Schachter J, West SK, Mabey D, Dawson CR, Bobo L, Bailey R, et al. Azithromycin in control of trachoma. Lancet. 1999; 354:630–5. https://doi.org/10.1016/S0140-6736(98)12387-5 PMID: 10466664
- Emerson PM, Lindsay SW, Alexander N, Bah M, Dibba S-M, Faal HB, et al. Role of flies and provision of latrines in trachoma control: cluster-randomised controlled trial. Lancet. 2004; 363:1093–8. <u>https:// doi.org/10.1016/S0140-6736(04)15891-1</u> PMID: 15064026
- Stoller NE, Gebre T, Ayele B, Zerihun M, Assefa Y, Habte D, et al. Efficacy of latrine promotion on emergence of infection with ocular Chlamydia trachomatis after mass antibiotic treatment: a cluster-randomized trial. Int Health. 2011; 3:75–84. https://doi.org/10.1016/j.inhe.2011.03.004 PMID: 21785663
- West S, Munoz B, Lynch M, Kayongoya A, Chilangwa Z, Mmbaga B, et al. Impact of face-washing on trachoma in Kongwa, Tanzania. Lancet. 2003; 345:155–8.
- Solomon AW, Holland MJ, Alexander ND, Massae PA, Aguirre A, Natividad-Sancho A, et al. Mass treatment with single-dose azithromycin for trachoma. N Engl J Med. 2004; 351:1962–71. https://doi.org/10. 1056/NEJMoa040979 PMID: 15525721
- Chidambaram JD, Alemayehu W, Melese M, Lakew T, Yi EH, House JI, et al. Effect of a single mass antibiotic distribution on the prevalence of infectious trachoma. JAMA. 2006; 295:1142–6. https://doi. org/10.1001/jama.295.10.1142 PMID: 16522834

- 11. World Health Organization. Trachoma control: a guide from programme managers. Geneva: World Health Organization; 2006 [cited 2017 Jul 2]. Available from: http://apps.who.int/iris/bitstream/handle/ 10665/43405/9241546905\_eng.pdf.
- Last A, Burr S, Alexander N, Harding-Esch E, Roberts CH, Nabicassa M, et al. Spatial clustering of high load ocular Chlamydia trachomatis infection in trachoma: a cross-sectional population-based study. Pathog Dis. 2017; 75:175–10. https://doi.org/10.1093/femspd/ftx050 PMID: 28472466
- Last AR, Burr SE, Harding-Esch E, Cassama E, Nabicassa M, Roberts CH, et al. The impact of a single round of community mass treatment with azithromycin on disease severity and ocular Chlamydia trachomatis load in treatment-naïve trachoma-endemic island communities in West Africa. Parasit Vectors. 2017; 10:624. https://doi.org/10.1186/s13071-017-2566-x PMID: 29282126
- 14. Chidambaram JD, Lee DC, Porco TC, Lietman TM. Mass antibiotics for trachoma and the Allee effect. Lancet Infect Dis. 2005; 5:194–6. https://doi.org/10.1016/S1473-3099(05)70032-3 PMID: 15792732
- 15. Allee WC. Principles of animal ecology. Philadelphia: Saunders Co; 1949.
- Lakew T, House J, Hong KC, Yi E, Alemayehu W, Melese M, et al. Reduction and return of infectious trachoma in severely affected communities in Ethiopia. PLoS Negl Trop Dis. 2009; 3:e376. https://doi. org/10.1371/journal.pntd.0000376 PMID: 19190781
- West SK, Munoz B, Mkocha H, Holland MJ, Aguirre A, Solomon AW, et al. Infection with Chlamydia trachomatis after mass treatment of a trachoma hyperendemic community in Tanzania: a longitudinal study. Lancet. 2005; 366:1296–300. https://doi.org/10.1016/S0140-6736(05)67529-0 PMID: 16214600
- Gebre T, Ayele B, Zerihun M, Genet A, Stoller NE, Zhou Z, et al. Comparison of annual versus twiceyearly mass azithromycin treatment for hyperendemic trachoma in Ethiopia: a cluster-randomised trial. Lancet. 2012; 379:143–51. https://doi.org/10.1016/S0140-6736(11)61515-8 PMID: 22192488
- Basilion EV, Kilima PM, Mecaskey JW. Simplification and improvement of height-based azithromycin treatment for paediatric trachoma. Trans R Soc Trop Med Hyg. 2005; 99:6–12. <u>https://doi.org/10.1016/j. trstmh.2004.01.014 PMID: 15550255</u>
- Emerson PM, Hooper PJ, Sarah V. Progress and projections in the program to eliminate trachoma. PLoS Negl Trop Dis. 2017; 11:e0005402. <u>https://doi.org/10.1371/journal.pntd.0005402</u> PMID: 28426814
- Diamant J, Benis R, Schachter J, Moncada J, Pang F, Jha HC, et al. Pooling of Chlamydia laboratory tests to determine the prevalence of ocular Chlamydia trachomatis infection. Ophthalmic Epidemiol. 2009; 8:109–17. https://doi.org/10.1076/opep.8.2.109.4156
- Melese M, Alemayehu W, Lakew T, Yi EH, House JI, Chidambaram JD, et al. Comparison of annual and biannual mass antibiotic administration for elimination of infectious trachoma. JAMA. 2008; 299:778–84. https://doi.org/10.1001/jama.299.7.778 PMID: 18285589
- Mabey DC, Downes RM, Downes B, Bailey RL, Dunn DT. The impact of medical services on trachoma in a Gambian village: antibiotics alone are not the answer. Ann Trop Paediatr. 1991; 11:295–300. PMID: 1719932
- Bowman RJ, Sillah A, van Dehn C, Goode VM, Muquit M, Johnson GJ, et al. Operational comparison of single-dose azithromycin and topical tetracycline for trachoma. Invest Ophthalmol Vis Sci. 2000; 41:4074–9. PMID: 11095598
- 25. Ngondi J, Gebre T, Shargie EB, Adamu L, Ejigsemahu Y, Teferi T, et al. Evaluation of three years of the SAFE strategy (Surgery, Antibiotics, Facial cleanliness and Environmental improvement) for trachoma control in five districts of Ethiopia hyperendemic for trachoma. Trans R Soc Trop Med Hyg. 2009; 103:1001–10. https://doi.org/10.1016/j.trstmh.2008.11.023 PMID: 19178920
- House JI, Ayele B, Porco TC, Zhou Z, Hong KC, Gebre T, et al. Assessment of herd protection against trachoma due to repeated mass antibiotic distributions: a cluster-randomised trial. Lancet. 2009; 373:1111–8. https://doi.org/10.1016/S0140-6736(09)60323-8 PMID: 19329003
- Liu F, Porco TC, Ray KJ, Bailey RL, Mkocha H, Munoz B, et al. Assessment of transmission in trachoma programs over time suggests no short-term loss of immunity. PLoS Negl Trop Dis. 2013; 7:e2303. https://doi.org/10.1371/journal.pntd.0002303 PMID: 23875038
- Lietman TM, Porco T, Dawson C, Blower S. Global elimination of trachoma: how frequently should we administer mass chemotherapy? Nat Med. 1999; 5:572–7. https://doi.org/10.1038/8451 PMID: 10229236
- Gaynor BD, Miao Y, Cevallos V, Jha H, Chaudary J, Bhatta R, et al. Eliminating trachoma in areas with limited disease. Emerg Infect Dis. 2003; 9:596–8. https://doi.org/10.3201/eid0905.020577 PMID: 12737745
- Stocks ME, Ogden S, Haddad D, Addiss DG, McGuire C, Freeman MC. Effect of water, sanitation, and hygiene on the prevention of trachoma: a systematic review and meta-analysis. PLoS Med. 2014; 11: e1001605. https://doi.org/10.1371/journal.pmed.1001605 PMID: 24586120

- Ngondi J, Gebre T, Shargie EB, Adamu L, Teferi T, Zerihun M, et al. Estimation of effects of community intervention with Antibiotics, Facial cleanliness, and Environmental improvement (A,F,E) in five districts of Ethiopia hyperendemic for trachoma. Br J Ophthalmol. 2010; 94:278–81. https://doi.org/10.1136/bjo. 2009.168260 PMID: 19897474
- Solomon AW, Mohammed Z, Massae PA, Shao JF, Foster A, Mabey DCW, et al. Impact of mass distribution of azithromycin on the antibiotic susceptibilities of ocular Chlamydia trachomatis. Antimicrob Agents Chemother. 2005; 49:4804–6. https://doi.org/10.1128/AAC.49.11.4804-4806.2005 PMID: 16251338
- Hong KC, Schachter J, Moncada J, Zhou Z, House J, Lietman TM. Lack of macrolide resistance in Chlamydia trachomatis after mass azithromycin distributions for trachoma. Emerg Infect Dis. 2009; 15:1088–90. https://doi.org/10.3201/eid1507.081563 PMID: 19624926
- Haug S, Lakew T, Habtemariam G, Alemayehu W, Cevallos V, Zhou Z, et al. The decline of pneumococcal resistance after cessation of mass antibiotic distributions for trachoma. Clin Infect Dis. 2010; 51:571–4. https://doi.org/10.1086/655697 PMID: 20649409
- Skalet AH, Cevallos V, Ayele B, Gebre T, Zhou Z, Jorgensen JH, et al. Antibiotic selection pressure and macrolide resistance in nasopharyngeal Streptococcus pneumoniae: a cluster-randomized clinical trial. PLoS Med. 2010; 7:e1000377. https://doi.org/10.1371/journal.pmed.1000377 PMID: 21179434
- Seidman JC, Coles CL, Silbergeld EK, Levens J, Mkocha H, Johnson LB, et al. Increased carriage of macrolide-resistant fecal E. coli following mass distribution of azithromycin for trachoma control. Int J Epidemiol. 2014; 43:1105–13. https://doi.org/10.1093/ije/dyu062 PMID: 24659584
- Seidman JC, Johnson LB, Levens J, Mkocha H, Munoz B, Silbergeld EK, et al. Longitudinal comparison of antibiotic resistance in diarrheagenic and non-pathogenic Escherichia coli from young Tanzanian children. Front Microbiol. 2016; 7:365–8. https://doi.org/10.3389/fmicb.2016.00365
- Ejere HO, Alhassan MB, Rabiu M. Face washing promotion for preventing active trachoma. Cochrane Database Syst Rev. 2015; 2:CD003659. <u>https://doi.org/10.1002/14651858.CD003659.pub4</u> PMID: 25697765
- 39. Harding-Esch EM, Sillah A, Edwards T, Burr SE, Hart JD, Joof H, et al. Mass treatment with azithromycin for trachoma: when is one round enough? Results from the PRET Trial in The Gambia. PLoS Negl Trop Dis. 2013; 7:e2115. https://doi.org/10.1371/journal.pntd.0002115 PMID: 23785525